Author’s response to reviews

Title: A Case of Bilateral Revision Total Knee Arthroplasty Using Distal Femoral Allograft-Prosthesis Composite and Femoral Head Allografting at the Tibial site with a Varus-valgus Constrained Prosthesis: Ten-year Follow up

Authors:
Sung Hyun Lee (kensin06@hanmail.net)
Sung Hyun Noh (nongoon@hanmail.net)
Keun Churl Chun (daystar30@nate.com)
Joung Kyue Han (hjk209037@hanmail.net)
churl hong chun (cch@wonkwang.ac.kr)

Version: 3 Date: 09 Feb 2018

Author’s response to reviews:

Thank you for giving us the opportunity to submit a revised version of our manuscript, now titled “Bilateral Revision Total Knee Arthroplasty Using Distal Femoral Allograft–Prosthesis Composite and Femoral Head Allografting at the Tibial site with a Varus-valgus Constrained Prosthesis: Ten-year Follow up.” The manuscript ID is BMSD-D-17-01202.

We have carefully considered all the reviewers’ comments and suggestions, and we have revised our paper accordingly. The comments were very helpful, and we appreciate the constructive feedback on our original submission. We think that the quality of the revised manuscript has improved considerably, and we hope that you agree.

We hope that the revised manuscript is now suitable for publication in your journal.

I look forward to your reply.
Yours sincerely,

Churl Hong Chun, MD, PhD

Professor and Fellowship Director

Department of Orthopedic Surgery, School of Medicine, Wonkwang University, Muwang-ro 895, Iksan, Jeollabuk-do, South Korea.

Tel: 82-63-850-1363

Fax: 82-63-852-9329

E-mail: cch@wku.ac.kr

1. We have noticed that the abstract in the manuscript and the submission system differ. Please note that the abstract should have the following headings: Background, Case presentation and conclusions.

   → Thank you for your comment. I changed my abstract.

2. Please change the "Introduction" heading to 'Background'.

   → Thank you for your comment. I changed to 'Background'.

3. Please change the "Discussion" heading to 'Discussion and conclusions'.

   → Thank you for your comment. I changed to 'Discussion and conclusions'.
4. Please provide a list of all the abbreviations used in the manuscript. This list should be placed just before the Declarations section. All abbreviations should still be defined in the text at first use.

→ Thank you for your comment. I have added content.

5. We note that the Declarations, present in your original manuscript, is not present in the latest revised version. Please note that all manuscripts must contain all the following sections under the heading 'Declarations'. The Declarations should follow the abbreviations section, and be before the References.

→ Thank you for your comment. I have added content.

We also noticed that in the original manuscript, the author "JKH" was missing from the Authors' contributions section. Please ensure to specify the contributions of ALL authors individually.

→ Thank you for your comment. I have added content.

6. Please remove the "revision note" and CARE checklist from the supplementary files.

→ Thank you for your comment. I removed the "revision note" and CARE checklist from the supplementary files.

7. Please revise your "timeline" in the supplementary files as it appears that some objects have moved.

→ Thank you for your comment. I revised "timeline" in the supplementary files
8. Please provide figure titles/legends under a separate heading of 'Figure Legends' after the References. If Figure titles/legends are within the main text of the manuscript, please move them.

→ Thank you for your comment. I have added content.

9. Please add a section "Additional files" (after the References/Figure legends) where you list the following information for each additional-supplementary file in the file inventory.

→ Thank you for your comment. I have added content.

10. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

→ Thank you for your comment. I have uploaded clean version.